Author's response to reviews

Title: Evaluation of Sexual Communication Message Strategies

Authors:

W Douglas Evans (wdevans@gwu.edu)
Kevin C Davis (kcdavis@rti.org)
Cindy Umanzor (cindy.umanzor@gmail.com)
Kajal Patel (pckkap@gwumc.edu)
Munziba Khan (munziba13@gmail.com)

Version: 4 Date: 14 May 2011

Author's response to reviews:

In this resubmission, we have responded to all the major changes called for in the editors' email of 5/13/2011.